Author's response to reviews

Title: Incidence of anogenital warts in Germany: a population-based cohort study

Authors:

   Angela A Kraut (kraut@bips.uni-bremen.de)
   Tania Schink (schink@bips.uni-bremen.de)
   Renate Schulze-Rath (rschulze-rath@spmsd.com)
   Rafael T Mikolajczyk (miko@bips.uni-bremen.de)
   Edeltraut Garbe (garbe@bips.uni-bremen.de)

Version: 5 Date: 9 December 2010

Author's response to reviews: see over
Dear Editors,

Please find attached our revised manuscript „Incidence of anogenital warts in Germany: a population-based cohort study”, MS: 1188685329427218.

In the method section we added a sentence clarifying that approval to use the data was granted from all statutory health insurances that contributed data to the study and by the Federal Ministry of Health.

We are looking forward to your reply.

Sincerely

[Signature]

Prof. Dr. med. Edeltraut Garbe